UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

SCHEDULE 13D

Under the Securities Exchange Act of 1934 (Amendment No.)*

LEJU HOLDINGS LIMITED

(Name of Issuer)

Ordinary Shares, par value \$0.001 per share

(Title of Class of Securities)

50187J108

(CUSIP Number)**

E-House (China) Enterprise Holdings Limited 11/F, Qiushi Building No. 785 Guangzhong Road Jiang'an District, Shanghai 200072 People's Republic of China Phone: 86 21 6086 8011

(Name, Address and Telephone Number of Person Authorized to Receive Notices and Communications)

with a copy to:

Z. Julie Gao, Esq. Shu Du, Esq. Skadden, Arps, Slate, Meagher & Flom LLP c/o 42/F, Edinburgh Tower, The Landmark 15 Queen's Road Central Hong Kong +852 3740-4700

November 4, 2020

(Date of Event Which Requires Filing of this Statement)

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D, and is filing this schedule because of §§240.13d-1(e), 240.13d-1(f) or 240.13d-1(g), check the following box. o

Note: Schedules filed in paper format shall include a signed original and five copies of the schedule, including all exhibits. See §240.13d-7 for other parties to whom copies are to be sent.

This statement on Schedule 13D (the "<u>Schedule 13D</u>") is an initial Schedule 13D filing on behalf of E-House (China) Enterprise Holdings Limited ("<u>E-House Enterprise</u>") with respect to the ordinary shares, par value \$0.001 per share (the "<u>Ordinary Shares</u>") of Leju Holdings Limited, an exempted company with limited liability incorporated under the laws of the Cayman Islands (the "<u>Issuer</u>").

The information required on the remainder of this cover page shall not be deemed to be "filed" for the purpose of Section 18 of the Securities Exchange Act of 1934 ("Act") or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

** This CUSIP number applies to the American depositary shares of the Issuer, each representing one Ordinary Share. No CUSIP has been assigned to the Ordinary Shares.

CUSIP No.	50187J108
-----------	-----------

1	NAMES OF REPORTING PERSONS			
	E-House (China) Enterprise Holdings Limited			
2			E BOX IF A MEMBER OF A GROUP	
	(a) o			
	(b) x			
3	SEC USE ONLY			
4	SOURCE OF FUN	DS (See Iı	nstructions)	
	00			
5		ISCLOSE	IRE OF LEGAL PROCEEDINGS IS REQUIRED	
	PURSUANT TO IT			
6	CITIZENSHIP OR	PLACE C	OF ORGANIZATION	
	Cayman Islands	7	SOLE VOTING POWER	
١ .	NUMBER OF	/	SOLE VOTING POWER	
	SHARES		76,401,247 Ordinary Shares	
	ENEFICIALLY	8	SHARED VOTING POWER	
	OWNED BY EACH			
1	REPORTING PERSON	9	0 SOLE DISPOSITIVE POWER	
	WITH	9	SOLE DISPOSITIVE POWER	
			76,401,247 Ordinary Shares	
		10	SHARED DISPOSITIVE POWER	
11	ACCDECATE AM	OUNT DI	0 ENEFICIALLY OWNED BY EACH REPORTING PERSON	
111	AGGREGATE AMOUNT BENEFICIALLY OWNED BY EACH REPORTING PERSON			
	76,401,247 Ordinary Shares			
12	CHECK IF THE AGGREGATE AMOUNT IN ROW (11) EXCLUDES CERTAIN SHARES (See Instructions)			
13	PERCENT OF CLASS REPRESENTED BY AMOUNT IN ROW (11)			
	FC 40V 1			
14	56.1% ¹	TING PER	SON (See Instructions)	
17	TYPE OF REPORTING PERSON (See Instructions)			
	СО			

This calculation is based on 136,303,688 Ordinary Shares outstanding as of November 4, 2020 (excluding 3,749,065 Ordinary Shares that had been issued to the depositary of the Issuer and reserved for future grants under the Issuer's share incentive plan).

Item 1. Security and Issuer

This Schedule 13D relates to the Ordinary Shares of the Issuer.

American depositary shares of the Issuer, each representing one Ordinary Share, are listed on the New York Stock Exchange under the symbol "LEJU."

The principal executive offices of the Issuer are located at Level G, Building G, No.8 Dongfeng South Road, Chaoyang District, Beijing 100016, People's Republic of China.

Item 2. Identity and Background

This Schedule 13D is being filed on behalf of E-House Enterprise (the "Reporting Person").

- (a)—(c), (f) E-House Enterprise is a company registered in the Cayman Islands with limited liability and listed on the main board of the Hong Kong Stock Exchange (stock code: 2048). E-House Enterprise is principally engaged in investment holding. The principal executive offices of E-House Enterprise are located at 11/F, Qiushi Building, No. 785 Guangzhong Road, Jiang'an District, Shanghai 200072, People's Republic of China. The name, business address, present principal occupation or employment and citizenship of each of the executive officers and directors of E-House Enterprise are set forth on Schedule A-1 hereto and are incorporated herein by reference.
- (d), (e) During the last five years, the Reporting Person or, to the best of its knowledge, any of the persons listed on Schedule A-1 hereto has been: (i) convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or (ii) a party to a civil proceeding of a judicial or administrative body of competent jurisdiction and as a result of such proceeding was or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.

Item 3. Source and Amount of Funds or Other Consideration

On November 4, 2020, E-House Enterprise completed the acquisition of an aggregate of 56.1% interest in the issued share capital of the Issuer. E-House Enterprise purchased (i) 51,925,996 Ordinary Shares of the Issuer from Mr. Xin Zhou and certain of his affiliated entities (the "Zhou Parties") by issuing to the Zhou Parties 166,918,440 of its ordinary shares ("E-House Shares"), and (ii) 24,475,251 Ordinary Shares of the Issuer from the Sina Corporation and an affiliated entity thereof (the "SINA Parties") by issuing to the SINA Parties 78,676,790 E-House Shares. Upon completion of these transactions, E-House Enterprise acquired the beneficial ownership of 76,401,247 Ordinary Shares of the Issuer.

Item 4. Purpose of Transaction

The Reporting Person reviews its investments on a continuing basis. Depending on overall market conditions, performance and prospects of the Issuer, subsequent developments affecting the Issuer, other investment opportunities available to the Reporting Person and other investment considerations, the Reporting Person may hold, vote, acquire or dispose of or otherwise deal with securities of the Issuer. Any of the foregoing actions may be effected at any time or from time to time, subject to applicable law. Except as set forth above, there are no plans or proposals of the type referred to in subparagraphs (a) through (j) of Item 4 of this Schedule 13D.

Item 5. Interest in Securities of the Issuer

(a)—(b) The responses of the Reporting Person to Rows (7) through (13) of the cover pages of this Schedule 13D are hereby incorporated by reference in this Item 5.

E-House Enterprise acquired the beneficial ownership of 76,401,247 Ordinary Shares, representing approximately 56.1% of the total issued and outstanding Ordinary Shares.

The information with respect to Ordinary Shares that may be deemed to be beneficially owned by each director and officer of E-House Enterprise is set forth on <u>Schedule A-2</u> hereto, which is incorporated herein by reference.

The percentage of Ordinary Shares beneficially owned by the Reporting Person is based on 136,303,688 Ordinary Shares outstanding as of November 4, 2020 (excluding 3,749,065 Ordinary Shares that had been issued to the depositary of the Issuer and reserved for future grants under the Issuer's share incentive plan).

- (c) Except as described in this Schedule 13D, neither the Reporting Person nor, to the best of its knowledge, any of the persons listed on Schedule A-1 hereto, has effected any transactions in the Ordinary Shares of the Issuer during the past 60 days.
 - (d) Not applicable.
 - (e) Not applicable.

ITEM 6. Contracts, Arrangements, Understandings or Relationships with respect to Securities of the Issuer.

The information set forth in Item 3 of this Schedule 13D is hereby incorporated by reference in this Item 6.

A copy of the equity transfer agreement between the Reporting Person and the Zhou Parties is attached hereto as Exhibit 7.01 and is incorporated by reference herein. A copy of the equity transfer agreement between the Reporting Person and the SINA Parties is attached hereto as Exhibit 7.02 and is incorporated by reference herein.

ITEM 7. Material to be Filed as Exhibits.

- Exhibit 7.01 Equity Transfer Agreement by and among E-House (China) Enterprise Holdings Limited, Zhou Xin, On Chance Inc., Kanrich Holdings Limited, Jun Heng Investment Limited and E-House (China) Holdings Limited, dated July 31, 2020
- Exhibit 7.02 Equity Transfer Agreement by and among E-House (China) Enterprise Holdings Limited, Sina Corporation and MemeStar Limited, dated July 31, 2020

SIGNATURE

After reasonable inquiry and to the best of its knowledge and belief, each of the undersigned certifies that the information set forth in this statement is true, complete and correct.

Date: November 20, 2020

E-House (China) Enterprise Holdings Limited

By: /s/ Liang Zhou

Name: Liang Zhou Title: Authorized Representative

SCHEDULE A-1

Directors and Executive Officers of E-House Enterprise

Name	Position with E-House Enterprise	Present Principal Occupation	Business Address	Citizenship
Zhou Xin	Chairman, Executive director	_	1	Chinese (Hong Kong)
Huang Canhao	Vice-Chairman, Executive director	_	1	Chinese (Hong Kong)
Cheng Li-Lan	Executive director	_	1	United States of America
Ding Zuyu	Chief Executive Officer, Executive director	_	1	Chinese
Li Silong	Non-executive director	General manager of the venture capital department at the headquarters of Country Garden Holdings Company Limited	2	Chinese
Zhang Hai	Non-executive director	Senior vice president of China Vanke Co., Ltd.	3	Chinese
Xie Mei	Non-executive director	Executive Director and chief executive of Overseas Chinese Town (Asia) Holdings limited	4	Chinese
Huang Haojun	Non-executive director	Assistant to the general manager (Shenzhen) of the investment management centre at the headquarters of China Evergrande Group	5	Chinese
Zhang Bang	Independent non-executive director	Chief corporate officer of Octave (Shanghai) Enterprise Management Company Limited	6	Chinese
Zhu Hongchao	Independent non-executive director	Head of office and senior partner at Shanghai United Law Firm	7	Chinese
Wang Liqun	Independent non-executive director	Chairman and founder of Stone Capital Co., Ltd.	8	Chinese
Li Jin	Independent non-executive director	Vice President of Guangzhou Shenba Information Technology Co., Ltd.	9	Chinese (Hong Kong)

- 1. The business address of Mr. Zhou Xin, Mr. Huang Canhao, Dr. Cheng Li-Lan and Dr. Ding Zuyu is 11/F, Qiushi Building, 383 Guangyan Road, Jing'an District, Shanghai 200072, China.
- 2. The business address of Mr. Li Silong is No.1 Country Garden Road, Beijiao, Shunde District, Foshan, Guangdong Province, China.
- 3. The business address of Mr. Zhang Hai is L9 T3 No.988 Shenchang Rd, Minhang District, Shanghai, China
- 4. The business address of Ms. Xie Mei is 3/F, Jacaranda IBC, OCT Harbour, Baishi Rd, Shenzhen, Guangdong Province, China.
- 5. The business address of Mr. Huang Haojun is 38/F, 1126 Haide 3rd Rd., Shenzhen, Guangdong Province, China.
- 6. The business address of Mr. Zhang Bang is 5/F, No. 20 Donghu Road, Xuhui District, Shanghai, China.
- 7. The business address of Mr. Zhu Hongchao is Room 1702, Bund Center, No.222 Yan'an East Road, Huangpu District, Shanghai, China.
- 8. The business address of Mr. Wang Liqun is Shanghai Stonecapital Ltd., No. 4, Lane 163 Maoming South Road, Huangpu District, Shanghai, China.
- 9. The business address of Mr. Li Jin is 9/F, T1, Poly International Plaza, No. 45 Dawangjing Business District, Chaoyang District, Beijing, China.

SCHEDULE A-2

Ordinary Shares Beneficially Owned by the Directors and Executive Officers of E-House Enterprise

	Ordinary Shares Benef	icially Owned
	Number	% ⁽¹⁾
Directors and Executive Officers		
71. V'.	220,000	0.20/
Zhou Xin	220,000	0.2%
Huang Canhao	87,500	0.1%
Cheng Li-Lan	394,835	0.3%
Ding Zuyu	21,667	0.0%
		515,0
Li Silong	-	_
Zhang Hai	_	_
Xie Mei	_	_
Huang Haojun	_	_
Zhang Bang	_	_
Zhu Hongchao	78,414	0.1%
Wang Liqun		_
2.4		
Li Jin	101,664	0.1%

⁽¹⁾ For each person included in this table, the percentage of beneficial ownership is calculated based on 136,303,688 Ordinary Shares outstanding as of November 4, 2020 (excluding 3,749,065 Ordinary Shares that had been issued to the depositary of the Issuer and reserved for future grants under the Issuer's share incentive plan)

31 July 2020

E-House (China) Enterprise Holdings Limited

and

Zhou Xin

On Chance Inc.

Kanrich Holdings Limited

Jun Heng Investment Limited

E-House (China) Holdings Limited

EQUITY TRANSFER AGREEMENT

CONTENTS

Clause	_		Page
1.	EOUITY	TRANSFER AND CONSIDERATION	2
2.	-	IONS TO CLOSING	2
3.	CLOSING		3
4.	REPRES	ENTATIONS, WARRANTIES AND UNDERTAKINGS	4
5.		OSING OBLIGATIONS	4
6.	EXPENS	ES	5
7.	INFORM	ATION AND REPORTING	5
8.	CONFID	ENTIALITY	6
9.	TERMIN	ATION	6
10.	ANNOUN	NCEMENTS	7
11.	NOTICES	S	7
12.	ASSIGNN	MENT	8
13.	FURTHE	R ASSURANCE	8
14.	GOVERN	NING LAW; JURISDICTION AND ARBITRATION	8
15.		ACTS (RIGHTS OF THIRD PARTIES) ORDINANCE	9
16.		AGRÈEMENT	9
17.	COUNTE	ERPARTS	9
18.	APPOINT	TMENT OF PROCESS AGENT	9
SCHED	OULE 1	SALE INTEREST AND CONSIDERATION SHARES	I
SCHED	OULE 2	CLOSING CONDITIONS	II
SCHED	OULE 3	CLOSING ARRANGEMENT	III
SCHED	OULE 4	REPRESENTATIONS AND WARRANTIES	IV
SCHED	OULE 5	DEFINITIONS	IX

THIS EQUITY TRANSFER AGREEMENT (this "Agreement") is made on 31 July 2020

BETWEEN:

- (1) E-House (China) Enterprise Holdings Limited, a company registered in the Cayman Islands with limited liability and listed on the main board of the Hong Kong Stock Exchange (stock code: 2048), whose registered office is at the offices of Maples Corporate Services Limited, PO Box 309, Ugland House, Grand Cayman KY1-1104, Cayman Islands and principal office in Hong Kong is at 18/F, Tesbury Centre, 28 Queen's Road East, Wan Chai, Hong Kong (the "Issuer");
- (2) Mr. Zhou Xin, an individual holding Hong Kong Identity Card Number R285049(3) and address of Flat C, 8/F, Tower 1, Sorrento, No.1 Austin Road West, Tsim Sha Tsui, Hong Kong ("Mr. Zhou").
- (3) On Chance Inc., a company incorporated under the laws of the British Virgin Islands whose registered office is at AMS Trustees Limited, Sea Meadow House, Blackburne Highway, P.O. Box 116, Road Town, Tortola, British Virgin Islands ("On Chance");
- (4) Kanrich Holdings Limited, a company incorporated under the laws of the British Virgin Islands whose registered office is at Commence Chambers, P.O. Box 2208, Road Town, Tortola, British Virgin Islands ("Kanrich");
- (5) Jun Heng Investment Limited, a company incorporated under the laws of the British Virgin Islands whose registered office is at Palm Grove House, P.O. Box 438, Road Town, Tortola, British Virgin Islands ("Jun Heng"); and
- (6) E-House (China) Holdings Limited, a company incorporated under the laws of the Cayman Islands whose registered office is at PO Box 309, Ugland House, Grand Cayman, KY1-1104, Cayman Islands ("E-House Holdings");
 - Mr. Zhou, On Chance, Kanrich, Jun Heng and E-House Holdings together the "**Sellers**". Each of the Issuer and the Sellers are individually referred to herein as a "**Party**", and together, as the "**Parties**".

Words and expressions used in this Agreement shall be interpreted in accordance with Schedule 5.

WHEREAS:

- (A) Each of On Chance, Kanrich, Jun Heng and E-House Holdings is directly or indirectly wholly-owned and controlled by Mr. Zhou.
- (B) On and subject to the terms and conditions set out in this Agreement, the Issuer proposes to purchase and the Sellers propose to sell in aggregate 49,686,192 ordinary shares and 2,239,804 American depositary shares ("ADSs", representing 2,239,804 ordinary shares), as detailed in Schedule 1, of Leju Holdings Limited ("Leju"), a company registered in the Cayman Islands and listed on the New York Stock Exchange (stock ticker: LEJU), representing in aggregate approximately 38.19% of the total equity interest in Leju (the "Sale Interest") as at the date of this Agreement.
- (C) The consideration for the sale and purchase of the Sale Interest shall be settled by the Issuer allotting and issuing 166,918,440 newly issued Shares (the "Consideration Shares"), as detailed in <u>Schedule 1</u>, to the Seller and/or the Eligible Nominee(s) (as defined below) on and subject to the terms and conditions set out in this Agreement.

(D) Upon Closing and subject to the terms of this Agreement, the Issuer will hold approximately 38.19% of the total equity interest in Leju as a result of this Agreement and on the assumption that there are no changes to the issued share capital of Leju.

It is hereby agreed as follows:

1. EQUITY TRANSFER AND CONSIDERATION

- 1.1 On and subject to the terms and conditions of this Agreement, the Issuer agrees to purchase and the Sellers agrees to sell the Sale Interest, being in aggregate 49,686,192 ordinary shares and 2,239,804 ADSs of Leju, representing approximately 38.19% of the total equity interest in Leju as at the date of this Agreement.
- 1.2 The consideration for the sale and purchase of the Sale Interest shall be the sum of US\$198,579,099 (equivalent to approximately HK\$1,538,988,015, being approximately US\$3.82 per ordinary share or ADS of Leju and an indicative valuation of Leju of US\$520 million), which shall be settled by the Issuer allotting and issuing Consideration Shares, being 166,918,440 newly issued Shares, to the Sellers and/or the Eligible Nominee(s) (as defined below), at the issue price of HK\$9.22 (equivalent to approximately US\$1.19), on and subject to the terms and conditions set out in this Agreement.
- 1.3 The Consideration Shares shall, when issued, be free and clear from all Encumbrance and shall, upon Closing, be credited as fully paid and rank *pari passu* in all respects with the other Shares in issue including the rights attaching to them.

2. CONDITIONS TO CLOSING

- 2.1 Subject to the provisions under this clause 2, Closing shall be conditional on the Closing Conditions being satisfied or waived in accordance with clause 2.2 with respect to all Parties.
- 2.2 The Closing Conditions set out in <u>Schedule 2</u> paragraph 3 may be waived, in whole or in part, at any time by the Sellers, by notice in writing to the Issuer and the Closing Conditions set out in <u>Schedule 2</u> paragraph 4 may be waived, in whole or in part, at any time by the Issuer, by notice in writing to the Sellers. For the avoidance of doubt, no other Closing Conditions may be waived by any Party.
- 2.3 The date on which all the Closing Conditions set out in <u>Schedule 2</u> have been satisfied (or waived in accordance with clause 2.2) shall be the "**Unconditional Date**". If the Unconditional Date has not occurred on or prior to ninety (90) days after the date of this Agreement (the "**Longstop Date**"), the Issuer may, at its sole discretion within five (5) Business Days after the Longstop Date:
 - (a) terminate this Agreement with immediate effect (other than the Surviving Provisions), by written notice to the Sellers, and (without prejudice to the rights and/or obligations of any Party in respect of any antecedent breach) the Parties shall be released and discharged from their respective obligations under this Agreement; or
 - (b) with the agreement of the Sellers, extend the Longstop Date to such later date as the Parties may agree (such date being not more than ninety (90) days after the Longstop Date) (such new date being the "**Extended Longstop Date**").
- If the Unconditional Date has not occurred on or prior to the Extended Longstop Date (if any), this Agreement shall terminate with immediate effect (other than the Surviving Provisions and (without prejudice to the rights and/or obligations of any Party in respect of any antecedent breach) the Parties shall be released and discharged from their respective obligations under this Agreement.

- 2.5 The Issuer shall use its reasonable endeavours to ensure that the Closing Conditions set out in <u>Schedule 2</u> (excluding <u>Schedule 2</u> paragraph 4) are fulfilled or satisfied as soon as reasonably practicable after the date of this Agreement and remain so up to or as of the Closing Date (as the case may be).
- 2.6 The Sellers shall use their reasonable endeavours to ensure that the Closing Conditions set out in <u>Schedule 2</u> (excluding <u>Schedule 2</u> paragraph 3) be fulfilled or satisfied as soon as reasonably practicable after the date of this Agreement and remain so up to or as of the Closing Date (as the case may be).

3. CLOSING

- 3.1 Subject to the satisfaction or waiver of the Closing Conditions in accordance with the provisions of this Agreement, Closing shall take place at the offices of Skadden, Arps, Slate, Meagher & Flom, at 42nd Floor, Edinburgh Tower, The Landmark, 15 Queen's Road Central, Hong Kong, on the Closing Date at 2:00 p.m. (Hong Kong time) or by such other method (including the electronic exchange of relevant documents and items) or at such other place and time as the Parties may agree in writing.
- 3.2 At Closing, each of the Parties shall deliver or perform (or ensure that there is delivered or performed) all those documents, items and actions respectively listed in relation to that Party in Schedule 3.
- 3.3 All documents and items delivered at Closing pursuant to clause 3.2, shall be held by the recipient to the order of the person delivering the same until such time as Closing shall be deemed to have taken place. Simultaneously with delivery of all documents and all items required to be delivered at such Closing pursuant to clause 3.2 (or waiver of the delivery of it by the person entitled to receive the relevant document or item), the documents and items delivered in accordance with Schedule 3 shall cease to be held to the order of the person delivering them, the actions required to take place in accordance with Schedule 3 shall occur and Closing shall be deemed to have taken place.
- 3.4 Each Seller shall be entitled to nominate one or more entities to take up its Consideration Shares upon Closing by giving written notice to the Issuer no later than three (3) Business Days before the Closing Date, provided that:
 - (a) such entities are majority owned and controlled by the Seller ("**Eligible Nominee(s)**");
 - (b) such nomination shall not affect the liabilities and obligations of the Seller under this Agreement; and
 - (c) each relevant Eligible Nominee enters into a lock-up undertaking in favour of the Issuer on terms substantially the same as those set out in clause 5.1 for the remainder of the Lock-up Periods.

The written notice referred to in this clause 3.4 shall contain the full name and legal address of each relevant Eligible Nominee and the number of Consideration Shares to be taken up by it, and the full name and address of the process agent in Hong Kong of such Eligible Nominee (if applicable).

4. REPRESENTATIONS, WARRANTIES AND UNDERTAKINGS

- 4.1 Subject to and except for the information in any announcement, circular or other release or communication released or published by the Issuer, the Issuer represents and warrants to the Sellers that the Issuer's Warranties set out in Part A of Schedule 4 are true, accurate and complete as at the date of this Agreement. The Issuer's Warranties, save for those that are expressed to be given only as of a specific date, shall be deemed to be repeated by the Issuer immediately prior to Closing by reference to the facts and circumstances then existing as if references in the Issuer's Warranties to the date of this Agreement were references to the Closing Date.
- 4.2 The Sellers represent and warrant to the Issuer that the Sellers' Warranties set out in Part B of Schedule 4 are true, accurate and complete as at the date of this Agreement. The Sellers' Warranties, save for those that are expressed to be given only as of a specific date, shall be deemed to be repeated by the Sellers for themselves and on behalf of any Eligible Nominees immediately prior to Closing by reference to the facts and circumstances then existing as if references in the Sellers' Warranties to the date of this Agreement were references to the Closing Date.
- 4.3 Each Warranty given pursuant to clauses 4.1 and 4.2 is a separate and independent statement and is not limited or otherwise affected by any other Warranty or by any other provision of this Agreement.
- 4.4 Each Party hereby undertakes to notify the other Parties in writing promptly if it becomes aware of any circumstance arising at any time after the date of this Agreement which would cause any Warranty (if the Warranties were repeated with reference to the facts and circumstances then existing) to become untrue or inaccurate or misleading if such Warranty were repeated at any time prior to Closing.

5. POST-CLOSING OBLIGATIONS

- 5.1 Subject to the provisions in clause 5.2, the Sellers undertake to the Issuer that it will not (and where applicable, will procure that its Affiliates will not), except with the prior written consent of the Issuer:
 - (a) offer, pledge, sell, contract to sell, grant any option or contract to purchase, purchase any option or contract to sell or otherwise dispose of, directly or indirectly, any Shares or any beneficial or economic interest therein,
 - (b) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the direct or indirect economic consequences of the ownership of any Shares; or
 - (c) create any Encumbrance over, either directly or indirectly, conditionally or unconditionally, any Shares or any beneficial or economic interest therein,

for the following periods

- (x) within twelve (12) months of the Closing Date (the "**Initial Lock-up Period**"); and
- (y) within twelve (12) months of the date of the conversion in full of the HK\$1,031,900,000 2% Convertible Note Due 2023 to be issued by the Issuer to Alibaba.com Hong Kong Limited (the "**Noteholder**"), to the extent that any such actions would result in the Sellers and its Affiliates holding in aggregate less than the interest held by the Noteholder and its Affiliates (the "**Conversion Lock-up Period**") (the Initial Lock-up Period and the Conversion Lock-up Period, together the "**Lock-up Periods**").

- 5.2 Notwithstanding the provisions of clause 5.1 and for the avoidance of doubt, the Sellers, and its Affiliates, as the case may be, shall, within the Lock-up Periods, be permitted:
 - (a) to pledge any Shares, provided that such pledge is in place at the date of this Agreement, or to renew or replace any such pledge; and
 - (b) to transfer up to 3% of the issued share capital of the Company as at the Closing Date to a trust or other legal person for the purposes of creating an equity incentive pool for the management of the Group.
- 5.3 The Sellers and their Affiliates undertake to the Issuer that they shall not acquire any voting rights in respect of any Shares if such acquisition would result in an obligation arising under the Takeovers Code to make a general offer for Shares.

6. EXPENSES

- 6.1 Save as otherwise expressly provided in this Agreement, each of the Issuer and the Sellers shall be liable for its own costs and expenses (including but not limited to the costs and expenses of their own legal and other professional advisers) incurred in connection with the transactions contemplated herein.
- 6.2 Without prejudice to clause 6.1, all stamp, issue, registration, documentary or other similar taxes and duties payable on or in connection with the allotment and issue of the Consideration Shares shall be paid by the Issuer.
- 6.3 The Issuer agrees to pay all fees, costs and expenses incurred in connection with the listing of the Consideration Shares on the Hong Kong Stock Exchange.
- 6.4 Without prejudice to clause 6.1, all stamp, issue, registration, documentary or other similar taxes and duties payable on or in connection with the transfer of the Sale Interest shall be paid by the Sellers.

7. INFORMATION AND REPORTING

7.1 The Sellers shall, and shall procure their Affiliates to, comply with Applicable Laws (including but not limited to the Listing Rules and the Takeovers Code) in respect of the transactions contemplated under this Agreement and supply such information in relation to itself or its Affiliates or any transferee or prospective transferee of the Consideration Shares as may be necessary for inclusion in the documents to be despatched or announcements to be issued by the Issuer in accordance with or pursuant the Listing Rules, the Takeovers Code or other Applicable Laws or as requested by any applicable regulatory authority including the Hong Kong Stock Exchange and the SFC, take responsibility for such information in relation to itself and its Affiliates and authorise the publication, despatch and/or release of such documents and announcements containing information provided by it or its Affiliates.

8. CONFIDENTIALITY

- 8.1 Each Party undertakes that it shall (and shall procure that its Affiliates shall, and where relevant, undertakes to procure that its officers, employees, agents, investment managers and professional and other advisers and those of any Affiliate (together its "Authorised Persons") shall) use its best endeavours to keep confidential at all times and not permit or cause the disclosure of any information (other than to its Authorised Persons) which it may have or acquire before or after the date of this Agreement relating to the provisions of, and negotiations leading to, this Agreement and the performance of the obligations thereunder (such information being "Confidential Information"). In performing its obligations under this clause 8.1, each Party shall apply confidentiality standards and procedures at least as stringent as those it applies generally in relation to its own confidential information.
- 8.2 Each of the Parties agrees that it shall not use Confidential Information for any purpose other than in relation to the proper performance of its obligations and exercise of its rights under this Agreement and the transactions contemplated hereunder.
- 8.3 Each Party shall use its reasonable endeavours to alert the other Party as soon as is reasonably practical after it becomes aware of any request from a third party for disclosure of any Confidential Information.
- 8.4 The obligation of confidentiality under clause 8.1 does not apply to:
 - (a) information which at the date of disclosure is within the public domain (otherwise than as a result of a breach of this clause 8);
 - (b) the disclosure of information to the extent required to be disclosed by law, regulation or any court, tribunal or regulatory authority; or
 - (c) any announcement made in accordance with the terms of clause 10.

9. TERMINATION

- 9.1 If at any time prior to Closing it becomes apparent that any of the Issuer's Warranties is, or any event occurs or matter arises which renders, any of the Issuer's Warranties untrue or incorrect in any material respect as at the date of this Agreement or at Closing, then, the Sellers may, in their sole and reasonable discretion, by notice in writing to the Issuer, terminate this Agreement with immediate effect and the provisions of clause 9.4 shall apply.
- 9.2 If at any time prior to Closing it becomes apparent that any of the Sellers' Warranties is, or any event occurs or matter arises which renders, any of the Sellers' Warranties untrue or incorrect in any material respect as at the date of this Agreement or at Closing, then the Issuer may, in its sole and reasonable discretion, by notice in writing to the Sellers, terminate this Agreement with immediate effect and the provisions of clause 9.4 shall apply.
- 9.3 Without prejudice to clauses 2.3, 2.4, 9.1 or 9.2, this Agreement may be terminated at any time prior to Closing by written agreement between the Parties.
- 9.4 If there is a termination pursuant to clauses 2.3, 2.4, 9.1, 9.2 or 9.3, this Agreement shall terminate between the Parties and neither Party shall have any claim against the other in respect of any matter or thing arising out of or in connection with this Agreement except (i) in relation to obligations, agreements and liabilities arising prior to such termination, (ii) in relation to any antecedent breach of any obligation under this Agreement and (iii) the continued application of the Surviving Provisions.

10. ANNOUNCEMENTS

- 10.1 Subject to clauses 10.2 and 10.3, no public announcement of any kind in connection with this Agreement or the transactions contemplated thereunder shall be made or issued by or on behalf of any Party or any of its Affiliates without the prior written consent of the other Party.
- 10.2 If any Party is required by law or by any stock exchange or by any governmental or regulatory authority to make any announcement in connection with this Agreement or the transactions contemplated thereunder (other than the announcement referred to in clause 10.3), the relevant Party shall, to the extent legally permissible, immediately notify the other Party and shall use all reasonable endeavours to accommodate the requests of such Party with respect to the terms and provisions of such announcement.
- 10.3 The Parties acknowledge and agree that the Issuer is required to publish or despatch announcements, circulars or other communications ("Communications") in relation to the subject matter contained herein following entry into this Agreement by the Parties in accordance with the Listing Rules. The Issuer will provide the Sellers with a draft of such Communications and permit the Sellers a reasonable period of time in all the circumstances to review and comment, and to use its reasonable efforts to accommodate all reasonable comments made by the Sellers, on the Communications prior to the publication or despatch of the Communications.

11. NOTICES

- Any notice in connection with this Agreement shall be in writing and delivered by hand, fax, e-mail, registered post or courier using an internationally recognised courier company. A notice shall be effective upon receipt and shall be deemed to have been received (i) at the time of delivery, if delivered by hand, registered post or courier, (ii) at the time of dispatch, if delivered by fax (subject to confirmation of uninterrupted transmission by the sender by a transmission report), or (iii) in the case of e-mail, at the time it leaves the email gateway of the sender, provided that, in either case, where delivery occurs outside Working Hours, notice shall be deemed to have been received at the start of Working Hours on the next following Business Day.
- 11.2 The addresses, fax numbers and the e-mail addresses of the Parties for the purpose of clause 11.1 are:

For the Issuer:

Address: 11/F, Qiu Shi Building, No.788 Guangzhou Road, Jing'an District,

Shanghai, China 200072

E-mail Address: zhouliang@ehousechina.com

For the attention of: Fred Zhou

With a copy to: Skadden, Arps, Slate, Meagher & Flom

Address: 42F, Edinburgh Tower, The Landmark, 15 Queen's Road Central, Hong

Kong

E-mail Address: christopher.betts@skadden.com

For the attention of: Mr. Christopher Betts

For the Sellers:

Address: 11/F, Yinli Building, No.788 Guangzhong Road, Jing'an District,

Shanghai, China 200072

Fax No: +86 21 61330888

E-mail Address: zhouxin@ehousechina.com

For the attention of: Zhou Xin

With a copy to: E-House (China) Holdings Limited

Address: 11/F, Yinli Building, No.788 Guangzhong Road, Jing'an District,

Shanghai, China 200072

Fax No: +86 21 61330888

E-mail Address: chenglilan@ehousechina.com

For the attention of: Cheng Li-Lan

11.3 All notices or formal communications under or in connection with this Agreement shall be in the English language or, if in any other language, accompanied by a translation into English. In the event of any conflict between the English text and the text in any other language, the English text shall prevail.

12. ASSIGNMENT

12.1 No Party may assign its rights under this Agreement without the prior written consent of the other Party except that the Sellers may make the nomination in accordance with clause 3.4.

13. FURTHER ASSURANCE

- 13.1 Each Party shall perform (or procure the performance of) all further acts and things and execute and deliver (or procure the execution and delivery of) such further documents, as may be required by law or as may be necessary or reasonably required by the Parties to implement and give effect to this Agreement.
- 13.2 Each Party shall procure that its Affiliates comply with all obligations under this Agreement which are expressed to apply to any such Affiliates.

14. GOVERNING LAW; JURISDICTION AND ARBITRATION

- 14.1 This Agreement is governed by and shall be construed in accordance with Hong Kong law.
- Any dispute, controversy, difference or claim arising out of, in connection with or relating to (in each such case, in any manner whatsoever) this Agreement (each a "**Dispute**"), including a Dispute as to the validity or existence of this Agreement, shall be resolved in accordance with the provisions of this clause 14.
 - (a) A Dispute shall be referred to and resolved finally by arbitration in Hong Kong administered by the Hong Kong International Arbitration Centre (the "**HKIAC**") in accordance with the Hong Kong International Arbitration Centre Administered Arbitration rules (the "**HKIAC Rules**") in force when the Notice of Arbitration is submitted in accordance with the HKIAC Rules, save as modified in clause 14.2(b) below.
 - (b) A Dispute shall be resolved by an arbitral tribunal consisting of three (3) arbitrators, one appointed by the Issuer, one appointed by the Sellers and the third appointed by the first two arbitrators. The arbitration (including arbitration proceedings and hearing) shall be conducted in English.

- (c) Any arbitral award shall be final and binding upon the Parties thereto and shall be enforceable in accordance with its terms. Neither of the Parties shall seek to commence any judicial proceeding with a view to appealing, reviewing or setting aside any arbitral award. All such rights of appeal or judicial review of any arbitral award as would otherwise be exercisable by the Parties are hereby excluded to the fullest extent permitted.
- (d) Any arbitral award may be enforced by filing as a judgment in any court of competent jurisdiction or by any other competent application or proceeding in any such court for the enforcement of the arbitral award, as the case may be.

15. CONTRACTS (RIGHTS OF THIRD PARTIES) ORDINANCE

15.1 A person who is not a party to this Agreement shall not have any rights under the Contracts (Rights of Third Parties) Ordinance (Chapter 623 of the Laws of Hong Kong) to enforce any terms of this Agreement and the Parties may amend, vary, waive, terminate or rescind this Agreement at any time and in any way without the consent of and without notice to any third party.

16. ENTIRE AGREEMENT

16.1 This Agreement, together with any agreements or documents referred to herein, sets out the entire agreement and understanding between the Parties with respect to its subject matter and supersedes all prior agreements, understandings, negotiations and discussions (whether oral or written) and all previous agreements in relation to the subject matter contained herein are hereby terminated and shall have no further force or effect.

17. COUNTERPARTS

17.1 This Agreement may be executed in any number of counterparts, and by each Party on separate counterparts. Each counterpart is an original, but all counterparts shall together constitute one and the same instrument. Delivery of a counterpart of this Agreement by e-mail attachment or telecopy shall be an effective mode of delivery.

18. APPOINTMENT OF PROCESS AGENT

- 18.1 Each of the Sellers (other than Mr. Zhou) hereby irrevocably appoint Jennifer Chan, of Room 1706, 17/F, Two Exchange Square, Central, Hong Kong, as their agent to accept service of process in Hong Kong in any legal action or proceedings arising out of this Agreement, service upon whom shall be deemed completed whether or not forwarded to or received by it.
- 18.2 The Sellers agree to inform the Issuer in writing of any change of address of their process agent or authorised representative within twenty (20) Business Days of such change.
- 18.3 If the process agent of the Sellers ceases to be able to act as such or to have an address in Hong Kong, each of the Sellers irrevocably agrees to promptly appoint a new process agent in Hong Kong acceptable to the Issuer and to deliver to the Issuer within ten (10) Business Days a copy of a written acceptance of appointment by the process agent.
- 18.4 Nothing in this Agreement shall affect the right to serve process in any other manner permitted by law.

In Witness Whereof this Agreement has been duly executed by the authorised representatives of the Parties on the date first above written.				
SIGNED by Zhou Xin for and on behalf of E-House (China) Enterprise Holdings Limited in the presence of: /s/ [WITNESS]) /s/ Zhou Xin))			

SIGNED by Zhou Xin	/s/ Zhou Xin
in the presence of: /s/ [WITNESS]	
SIGNED by Zhou Xin for and on behalf of On Chance Inc.	/s/ Zhou Xin
in the presence of: /s/ [WITNESS]	
SIGNED by Zhou Xin for and on behalf of Kanrich Holdings Limited	/s/ Zhou Xin
in the presence of: /s/ [WITNESS]	
SIGNED by Zhou Xin for and on behalf of Jun Heng Investment Limited	/s/ Zhou Xin
in the presence of: /s/ [WITNESS]	
SIGNED by Zhou Xin for and on behalf of E-House (China) Holdings Limited in the presence of: /c//WHITNESS!	/s/ Zhou Xin
in the presence of: /s/ [WITNESS]	

SCHEDULE 1 SALE INTEREST AND CONSIDERATION SHARES

Seller	Ordinary shares in Leju	ADSs in Leju	Consideration Shares
Mr. Zhou Xin	370,833		
On Chance	203,240	2,058,879	_
Kanrich	889,506	_	_
Jun Heng	483,250	_	_
E-House Holdings	47,739,363	180,925	166,918,440
	i		

SCHEDULE 2 CLOSING CONDITIONS

- 1. The approval of this Agreement and the transactions and other arrangement (as necessary) contemplated hereunder, including the acquisition the Sale Interest and allotment and issue of the Consideration Shares, by the requisite majority of shareholders or independent shareholders of the Issuer (as appropriate) at an EGM as required under Applicable Laws (including the Takeovers Code and the Listing Rules).
- 2. The granting of the approval for the listing of, and permission to deal in, the Consideration Shares by the Hong Kong Stock Exchange.
- 3. The Issuer's Warranties set out in Schedule 4 Part A continuing to be true, accurate and complete in all material respects as of the Closing Date.
- 4. The Sellers' Warranties set out in Schedule 4 Part B continuing to be true, accurate and complete in all material respects as of the Closing Date.
- 5. The due completion of the equity transfer agreement of 31 July 2020 entered into between the Issuer, SINA Corporation and MemeStar Limited in respect of the acquisition by the Issuer of ordinary shares and ADSs in Leju from SINA Corporation and MemeStar Limited and the issuance of Consideration Shares by the Issuer to MemeStar Limited.
- 6. (A) no obligation having arisen for any of the Sellers or any party acting in concert with any of them to make a mandatory cash offer required to be made pursuant to Rule 26.1 of the Takeovers Code in respect of Shares, as a consequence of the transactions contemplated under this Agreement, or (B) in the event of any such obligation having arisen, a Whitewash Waiver having been granted by the Executive in accordance with the Takeovers Code.

SCHEDULE 3 CLOSING ARRANGEMENT

Part A Issuer Obligations

At Closing or otherwise within the time period specifically stated below, the Issuer shall perform all (but not part only unless the Sellers so agree) of the following:

- 1. issue and allot the Consideration Shares to the Sellers or the relevant Eligible Nominee(s), as the case may be, credited as fully-paid, and procure the entry of the Sellers or the relevant Eligible Nominee(s), as the case may be, in the register of members of the Issuer as the owner of the relevant Consideration Shares; and
- 2. deliver or ensure that there is delivered to the Sellers:
 - (a) (i) definitive certificates in respect of the relevant Consideration Shares in favour of the Sellers and/or its Eligible Nominee(s); or (ii) deposit the certificates into the account of the relevant CCASS Participant with whom the Seller has accounts in accordance with the Seller's instructions;
 - (b) a certified copy or extract of the resolutions of the Board with respect to approval of (i) the execution of this Agreement and the performance of the Issuer's obligations hereunder; and (ii) the issue of Consideration Shares to the Sellers or the relevant Eligible Nominee(s), as the case may be, in accordance with this Agreement; and
 - (c) a certificate signed by a duly authorised officer of the Issuer certifying that the matters in Schedule 2 paragraph 3 are true and correct to the Sellers.

Part B Seller's Obligations

At Closing, the Sellers shall:

- 1. deliver or ensure that there is delivered to the Issuer:
 - (a) a duly executed share transfer form in respect of the relevant Sale Interest in favour of the Issuer together with the share certificates relating to such Sale Interest (or an indemnity in respect of any lost unavailable share certificates in the agreed terms);
 - (b) a certified copy or extract of the resolutions of the board of the Sellers with respect to approval of the execution of this Agreement and the performance of the Seller's respective obligations hereunder; and
 - (c) a certificate signed by a duly authorised officer of the Sellers certifying that the matters in Schedule 2 paragraph 4 are true and correct for itself and behalf of any Eligible Nominees to the Issuer.

SCHEDULE 4 REPRESENTATIONS AND WARRANTIES

Part A Issuer's Warranties

The Issuer represents and warrants to the Sellers that:

- 1. **Incorporation and Subsidiaries:** each member of the Group is a company duly incorporated and validly existing and having its capital fully paid up under the laws of its jurisdiction of incorporation, is in compliance with its constitutional documents, and all laws, regulations and corporate governance requirements to which it is subject, is not in liquidation or receivership, has full power and authority to own its properties and to conduct its business and is lawfully qualified and licensed to do business in those jurisdictions in which business is conducted by it, and each of the foregoing statements in this paragraph is true with respect to each of the branches established by a member of the Group;
- 2. **Validity of Agreement:** this Agreement has, subject to the satisfaction of the Closing Conditions, been duly authorised, executed and delivered by the Issuer and constitutes valid and legally binding obligations of the Issuer, enforceable against the Issuer in accordance with its terms;
- 3. **Consideration Shares:** the Consideration Shares when issued and delivered:
 - (a) will be duly and validly issued and fully-paid and non-assessable;
 - (b) will rank *pari passu* with, and carry the same rights in all respects as the other Shares in issue and shall be entitled to all dividends and other distributions declared, paid or made thereon the record date of which falls after the Closing Date;
 - (c) will not be subject to any kind of pre-emptive rights;
 - (d) will be freely transferable, free and clear of all liens, charges, encumbrances, security interests or claims of third parties and will not be subject to calls for further funds; and
 - (e) will be duly listed, and admitted to trading, on the Main Board of the Hong Kong Stock Exchange;
- 4. **Restrictions:** save as otherwise provided for under this Agreement and in the Articles:
 - (a) there are no restrictions on transfers of the Consideration Shares; and
 - (b) there are no restrictions on the voting or transfer of any of the Consideration Shares or payments of dividends with respect to the Consideration Shares pursuant to the Issuer's constitutional documents, or pursuant to any agreement or other instrument to which the Issuer is a party or by which it is bound;
- 5. **Listing**: all of the issued Shares have been duly listed on the Hong Kong Stock Exchange;
- 6. **Consents:** save for the matters set out in the Closing Conditions, no consent, clearance, approval, authorisation, order, registration or qualification of or with any court, governmental agency or regulatory or self-regulatory body having jurisdiction over the Issuer is required and no other action or thing is required to be taken, fulfilled or done (including without limitation the obtaining of any consent or licence or the making of any filing or registration) for the execution and delivery by the Issuer of the Agreement, the issue of the Consideration Shares to the Sellers, the carrying out of the other transactions contemplated by this Agreement except for all of those which have been, or will prior to the Closing Date be, obtained and are, or will on the Closing Date be, in full force and effect. The Issuer has no knowledge of any facts that would reasonably lead to delisting or suspension of the Shares in the foreseeable future and the issuance by the Issuer of the Consideration Shares shall not have the effect of delisting or suspending the Shares from the Hong Kong Stock Exchange, in each case save with respect to the matters contained in clause 5.2 of this Agreement;

- 7. **Compliance:** the execution, delivery and performance of the Agreement, the issue of the Consideration Shares, the carrying out of the other transactions contemplated by the Agreement do not and will not:
 - (a) conflict with or result in a breach of any of the terms or provisions of, or constitute a default (nor has any event occurred which, with the giving of notice and/or the passage of time and/or the fulfilment of any other requirement would result in a default) by the Issuer or any other Group Company under, (a) the documents constituting the Issuer, or (b) any indenture, contract, lease, mortgage, deed of trust, note agreement, loan agreement or other agreement, obligation, condition, covenant or instrument to which the Issuer or any other Group Company is a party or by which any of their respective assets are bound; or
 - (b) infringe any existing applicable law, rule, regulation, judgment, order, authorisation or decree of any government, governmental or regulatory body or court, domestic or foreign, having jurisdiction over the Issuer or any other Group Company or any of their respective assets; or
 - (c) infringe the Listing Rules;
- 8. **No Suspension:** since the listing of the Shares on the Main Board of the Hong Kong Stock Exchange on 20 July 2018, and save for any trading halt required under the Listing Rules or the Takeovers Code in connection with the transactions contemplated herein, (i) the Shares have been listed on the Hong Kong Stock Exchange, (ii) trading in the Shares has not been suspended by the Hong Kong Stock Exchange since the listing on the Hong Kong Stock Exchange, and (iii) the Issuer has received no communication, written or oral, from the Hong Kong Stock Exchange regarding the suspension or delisting of the Shares from the Hong Kong Stock Exchange;
- 9. **Encumbrances:** no encumbrancer has taken possession of, and no receiver has been appointed over, the whole or any substantial part of the assets or undertaking of the Issuer or any other Group Company;
- 10. **Bankruptcy:** neither the Issuer nor any of its Subsidiaries is insolvent or bankrupt or unable to pay its debts, has not stopped, suspended or threatened (through actions of its board of directors or otherwise) to stop or suspend payment of all or any part of (or of a particular type of) its debts, has not proposed or made any agreement for the deferral, rescheduling or other readjustment of all of (or of a particular type of) its debts (or of any part which it will or might otherwise be unable to pay when due), has not proposed or made a general assignment or an arrangement or composition with or for the benefit of the relevant creditors in respect of any such debts and a moratorium has not been agreed or declared in respect of or affecting all or any part of (or of a particular type of) its debts; and
- 11. **Bankruptcy Proceedings:** no order has been made and no effective resolution has been passed for the winding-up or dissolution or administration or receivership of the Issuer or any of the Group Companies, and neither the Issuer nor any of the Group Companies has threatened (through actions of its board of directors or otherwise) to cease to carry on all or a material part of its business or operations except to the extent that it has made disposals of assets or shares in the ordinary course of its business for fair value.

Part B Seller's Warranties

The Sellers represents and warrants to the Issuer that:

- 1. with respect to itself and any Eligible Nominees:
 - (a) it is a company duly incorporated and organised and validly existing under the laws of its jurisdiction of incorporation or organisation, is not in liquidation or receivership and has full power and authority to own its properties and to conduct its business:
 - (b) it has power under its constitutional documents to subscribe for the Consideration Shares and hold the Consideration Shares upon the terms set out herein;
 - (c) it has full power and authority to enter into this Agreement and to perform its obligations hereunder;
 - (d) this Agreement has been duly authorised, executed and delivered by the Sellers and constitutes valid and legally binding obligations of the Sellers, enforceable in accordance with their respective terms; and
 - (e) the particulars and information relating to it as set out in this Agreement are true, correct and accurate and any information provided to the Issuer for the purposes of any Communications was, or will be, true, correct and accurate.
- 2. **Sale Interest:** the Sale Interest:
 - (a) are beneficially owned by the respective Sellers;
 - (b) when delivered:
 - (i) will be duly and validly issued and fully-paid and non-assessable;
 - (ii) will rank pari passu with, and carry the same rights in all respects as the other Leju ordinary shares and/or Leju ADSs (as the case may be) in issue and shall be entitled to all dividends and other distributions declared, paid or made thereon the record date of which falls after the Closing Date;
 - (iii) will not be subject to any kind of pre-emptive rights;
 - (iv) will be freely transferable, free and clear of all liens, charges, encumbrances, security interests or claims of third parties and will not be subject to calls for further funds; and
 - (c) are duly listed, and admitted to trading, on the Nasdaq Global Select Market;
- 3. **Restrictions:** save as otherwise provided for under this Agreement and in the articles of the Sellers:
 - (a) there are no restrictions on transfers of the Sale Interest; and
 - (b) there are no restrictions on the voting or transfer of any of the Sale Interest or payments of dividends with respect to the Sale Interest pursuant to Leju's constitutional documents, or pursuant to any agreement or other instrument to which any Seller is a party or by which it is bound;

- 4. **Listing**: all of the Sale Interest have been duly listed on the Nasdag Global Select Market.
- 5. **Leju Accounts:** in respect of the Leju's audited financial statements (comprising the consolidated statement of financial position as at 31 December 2019 and audited consolidated statement of profit or loss and other comprehensive income) for the year ended 31 December 2019) (the "**Leju Accounts**"):
 - (a) Other than as disclosed in the Leju Accounts, no unissued share capital of any member of Leju or its subsidiaries is under any option or agreed conditionally or unconditionally to be put under any option and no person has an outstanding warrant, pre-emptive right or any other right of any description to require shares to be allotted or issued by any of Leju or its subsidiaries.
 - (b) The Leju Accounts:
 - (i) were prepared in accordance with all Applicable Laws and the Generally Accepted Accounting Principles (United States) ("US GAAP") at the time they were prepared and on a consistent basis;
 - (ii) are true and accurate and give a true and fair view of the state of affairs and financial and trading positions of Leju and its subsidiaries as at 31 December 2019 and of the results of Leju and its subsidiaries for the financial period ended on that date;
 - (iii) include adequate provision for all Taxation relating to any period on or before 31 December 2019;
 - (iv) are not adversely affected by any unusual, exceptional, extraordinary or non-recurring items which are not expressly disclosed as such in the Leju Accounts; and
 - (v) make appropriate provision or reserve for, or disclose, all liabilities (including contingent and disputed liabilities) and all capital commitments of the Leju Group as at 31 December 2019, indicate clearly which of those liabilities are not usually provided for or reserved, and make adequate provision or reserve for all bad and doubtful debts.
 - (c) Since 31 December 2019 and save as disclosed in any public announcements or filings by Leju:
 - (i) apart from the dividends provided for in the Leju Accounts, no dividend or other distribution has been declared, paid or made by Leju;
 - (ii) the operation of Leju and its subsidiaries has been carried on in the ordinary and usual course;
 - (iii) no share or loan capital or other similar interest has been agreed or agreed to be issued by Leiu; and
 - (iv) Leju and its subsidiaries has not acquired or disposed of or agreed to acquire or dispose of any business or any material asset of a value exceeding HK\$100,000,000.00 other than in the ordinary course of business;
 - (v) other than in the ordinary course of business, Leju and its subsidiaries has not repaid any borrowing or indebtedness in advance of its stated maturity; and

- (vi) other than in the ordinary and usual course of business, no debtor of Leju and its subsidiaries has been released on terms that it pays less than the book value of its debt.
- (d) Leju and its subsidiaries are in compliance with all of its obligations under any outstanding guarantees or contingent payment obligations as described in the Leju Accounts.

SCHEDULE 5 DEFINITIONS

- The following terms and expressions used in this Agreement, unless the context otherwise requires, shall have the following meanings:
 - "Affiliate" of any specified person means any other person directly or indirectly Controlling or Controlled by or under direct or indirect common Control with such specified person;
 - "Applicable Laws" means, with respect to any person, any laws, rules, regulations, directives, treaties, decrees or orders of any governmental or regulatory authority that are applicable to and binding on such person and without limitation to the foregoing, shall, in respect of the Issuer, include the Listing Rules and the Takeovers Code;
 - "Articles" means the Issuer's articles of association from time to time;
 - "Authorised Persons" has the meaning given to it in clause 8.1;
 - "Board" means the board of directors of the Issuer;
 - "Business Day" means a day (other than a Saturday or Sunday or public holiday and any other day on which a tropical cyclone warning no. 8 or above or a "black" rain warning signal is hoisted in Hong Kong) on which commercial banks are open for business in the city in which the specified office of the registrar is located and in Hong Kong, and "Business Days" shall be construed accordingly;
 - "CCASS" means the Central Clearing and Settlement System established and operated by Hong Kong Securities Clearing Company Limited;
 - "CCASS Participant" means, as regards any Shares deposited with CCASS, the person admitted by HKSCC as a Participant (as defined in the General Rules of CCASS, as amended, issued by HKSCC);
 - "Closing" means the completion of the acquisition of the Sale Interest, as the case may be, in accordance with clause 3;
 - "Closing Conditions" means the conditions to Closing as set out in Schedule 2;
 - "Closing Date" means the date of Closing, which shall be a date not later than the fifth (5th) Business Day after the Unconditional Date, or such other date as the applicable Parties may agree in writing;
 - "Communications" has the meaning given to it in clause 10.3;
 - "Confidential Information" has the meaning given to it in clause 8.1;
 - "Consideration Shares" has the meaning given to it in Recital (B);
 - "Control" means the power to direct the management and policy of the entity, whether by virtue of voting shares, through contracts or by other means. In each case either directly or indirectly and "Controlled" and "Controlling" shall be construed accordingly;
 - "Director" means a director of the Issuer;
 - "Dispute" has the meaning given to it in clause 14.2;
 - "EGM" means the special general meeting of the Issuer to be held to approve, among other things, the acquisition of the Sale Interest and the issue of the Consideration Shares (as required under the Listing Rules and the Takeovers Code);

- "Eligible Nominee(s)" has the meaning given to it in clause 3.4(a);
- "Encumbrance" means any interest or equity (including any retention of title, right to acquire, option or right of pre-emption) or any mortgage, charge, pledge, lien, claim or assignment or any other encumbrance, priority or security interest or arrangement of whatsoever nature;
- "Executive" means the Executive Director of the Corporate Finance Division of the Securities and Futures Commission of Hong Kong, or any delegate of the Executive;
- "**Extended Longstop Date**" has the meaning given to it in clause 2.3(b);
- "Fairly Disclosed" means fairly and adequately disclosed in a manner and with sufficient details to enable the Sellers to reasonably identify and understand the nature and scope of the matter being disclosed to qualify the Issuer's Warranties;
- "Group" means the Issuer and together with its Subsidiaries from time to time;
- "Group Companies" means members of the Group, and "Group Company" means any one of them;
- "HK\$" means the legal currency of Hong Kong;
- "HKIAC" has the meaning given to it in clause 14.2(a);
- "HKIAC Rules" has the meaning given to it in clause 14.2(a);
- "HKSCC" means Hong Kong Securities Clearing Company Limited, a wholly-owned subsidiary of Hong Kong Exchanges and Clearing Limited;
- "Hong Kong" means the Hong Kong Special Administrative Region of the People's Republic of China;
- "Hong Kong Stock Exchange" means The Stock Exchange of Hong Kong Limited;
- "Issuer's Warranties" means the warranties set out in Schedule 4 Part A;
- "Listing Rules" means the Rules Governing the Listing of Securities on The Stock Exchange of Hong Kong Limited;
- "Lock-up Periods" has the meaning given to it in clause 5.1;
- "Longstop Date" has the meaning given to it in clause 2.3;
- "SFC" means the Securities and Futures Commission of Hong Kong;
- "Shares" means the ordinary shares with a par value of USD0.00001 each in the share capital of the Issuer;
- "Seller's Warranties" means the warranties set out in Schedule 4 Part B;
- "Subsidiary" includes, in relation to any person: (i) any company or business entity of which that person owns or controls (either directly or through one or more other subsidiaries) more than 50 per cent. of the issued share capital or other ownership interest having ordinary voting power to elect the directors, managers or trustees of such company or business entity; (ii) any company or business entity of which that person owns or controls (either directly or through one or more other subsidiaries) not more than 50 per cent. of the issued share capital or other ownership interest having ordinary voting power to elect directors, managers or trustees of such company or business entity but effectively controls (either directly or through one or more other subsidiaries) the management of the direction of business operations of such company or business entity; and (iii) any company or business entity which at any time has its accounts consolidated with those of that person or which, under Hong Kong law or any other applicable law regulations or the or such other applicably generally accepted accounting principles from time to time, should have its accounts consolidated with those of that person;

"Surviving Provisions" means clauses 1.3, 8, 11 and 14;

"Takeovers Code" means the Codes on Takeovers and Mergers and Share Buy-backs;

"Taxation" or "Tax" means income tax, corporation tax, capital gains tax, transfer taxes (including stamp duties, real estate transfer taxes, registration fees and other taxes of a similar nature), value added tax, customs duties, excise duties, national insurance and other similar social security contributions, and any other taxes, duties or withholdings corresponding to, similar to, replaced by or replacing any of them together with any interest, penalty or fine in connection with any such taxation and regardless of whether any such taxes, duties, withholdings, interest, penalties or fines are chargeable directly or primarily against or attributable directly or primarily to the Group or any other person and of whether any amount in respect of any of them is recoverable from any other person;

"Unconditional Date" has the meaning given to it in clause 2.3;

"USD" means United States dollars, the lawful currency of the United States;

"Warranties" mean the Issuer's Warranties and Seller's Warranties as set out in Schedule 4;

"Whitewash Waiver" means a waiver granted by the Executive pursuant to Note 1 of the Notes on dispensations from Rule 26 of the Takeovers Code from an obligation (which may otherwise arise as a result of Closing) on the part of the Sellers or any party acting in concert with any of them (within the meaning of the Takeovers Code) to make a general offer for the issued securities of the Company;

"Working Hour" means 9.30 a.m. to 5.30 p.m. on a Business Day in the place of receipt of a notice.

2. In this Agreement:

- (a) words denoting the singular shall include the plural and vice versa;
- (b) words denoting one gender shall include each gender and all genders;
- (c) references to persons shall be deemed to include references to natural persons, to firms, to partnerships, to bodies corporate, to undertakings, to associations, to organisations, to trusts, to trustees, to legal representatives, to governments (or any department or agency thereof) or to any other entity howsoever designated or constituted (in each case, whether or not having separate legal personality), but references to individuals shall be deemed to be references to natural persons only;
- (d) a reference to this Agreement includes the Schedules to it, each of which forms part of this Agreement for all purposes;
- (e) references to clauses and Schedules are, unless stated otherwise, references to clauses and schedules of this Agreement;
- (f) references to paragraphs are, unless expressly provided otherwise, references to paragraphs of the Schedule in which the references appear;

- (g) the headings are inserted for convenience only and will not affect the construction of this Agreement;
- (h) any reference to an enactment or a statutory provision is a reference to it as it may have been or may from time to time be, amended, modified, consolidated or re-enacted;
- (i) the terms "hereof" and "hereunder" (and any other similar expressions) refer to this Agreement and not to any particular clause or other portion hereof and include any agreement supplemental hereto; and
- (j) any statement in this Agreement qualified by the expression "to the best knowledge of the Issuer or any member of the Group" or "so far as the Issuer is aware" or any similar expression shall be a reference to the actual knowledge of the Directors and the members of the senior management team of the Issuer and shall be deemed to include an additional statement that it has been made after due and careful enquiry by such persons.

Execution Version

31 July 2020

E-House (China) Enterprise Holdings Limited

and

Sina Corporation

and

MemeStar Limited

EQUITY TRANSFER AGREEMENT

CONTENTS

CLAUSE		PAGE
1.	EQUITY TRANSFER AND CONSIDERATION	1
2.	CONDITIONS TO CLOSING	2
3.	CLOSING	2
4.	REPRESENTATIONS, WARRANTIES AND UNDERTAKINGS	3
5.	EXPENSES	4
6.	INFORMATION AND REPORTING	4
7.	CONFIDENTIALITY	4
8.	TERMINATION	5
9.	ANNOUNCEMENTS	5
10.	NOTICES	6
11.	ASSIGNMENT	6
12.	FURTHER ASSURANCE	7
13.	GOVERNING LAW; JURISDICTION AND ARBITRATION	7
14.	CONTRACTS (RIGHTS OF THIRD PARTIES) ORDINANCE	7
15.	ENTIRE AGREEMENT	7
16.	COUNTERPARTS	8
17.	APPOINTMENT OF PROCESS AGENT	8
SCHEDU	ULE 1 SALE INTEREST AND CONSIDERATION SHARES	I
SCHEDU	ULE 2 CLOSING CONDITIONS	II
SCHEDU	ULE 3 CLOSING ARRANGEMENT	III
SCHEDU	ULE 4 REPRESENTATIONS AND WARRANTIES	IV
SCHEDU	ULE 5 DEFINITIONS	VIII

THIS EQUITY TRANSFER AGREEMENT (this "Agreement") is made on 31 July 2020

BETWEEN:

- E-House (China) Enterprise Holdings Limited, a company registered in the Cayman Islands with limited liability and listed on the main board of the Hong Kong Stock Exchange (stock code: 2048), whose registered office is at the offices of Maples Corporate Services Limited, PO Box 309, Ugland House, Grand Cayman KY1-1104, Cayman Islands and principal office in Hong Kong is at 18/F, Tesbury Centre, 28 Queen's Road East, Wan Chai, Hong Kong (the "Issuer");
- (2) Sina Corporation, a company incorporated in the Cayman Islands and listed on the Nasdaq Global Select Market (stock ticker: SINA), whose registered office is at the offices of Maples Corporate Services Limited, Ugland House, P.O. Box 309, Grand Cayman, KY1-1104, Cayman Islands ("Sina"); and
- (3) MemeStar Limited, a company incorporated under the laws of the British Virgin Islands whose registered office is at Offshore Incorporations Limited, P O. Box 957, Offshore Incorporations Centre, Road Town, Tortola, British Virgin Islands ("MemeStar");

Sina and MemeStar together the "Sellers". Each of the Issuer and the Sellers are individually referred to herein as a "Party", and together, as the "Parties".

Words and expressions used in this Agreement shall be interpreted in accordance with Schedule 5.

WHEREAS:

- (A) MemeStar is wholly-owned and controlled by Sina.
- (B) On and subject to the terms and conditions set out in this Agreement, the Issuer proposes to purchase and the Sellers propose to sell in aggregate 24,438,564 ordinary shares and 36,687 American depositary shares ("ADSs", representing 36,687 ordinary shares), as detailed in Schedule 1, of Leju Holdings Limited ("Leju"), a company registered in the Cayman Islands and listed on the New York Stock Exchange (stock ticker: LEJU), representing in aggregate approximately 18.00% of the total equity interest in Leju (the "Sale Interest") as at the date of this Agreement.
- (C) The consideration for the sale and purchase of the Sale Interest shall be settled by the Issuer allotting and issuing 78,676,790 newly issued Shares (the "Consideration Shares"), as detailed in <u>Schedule 1</u>, to the Sellers and/or the Eligible Nominee(s) (as defined below) on and subject to the terms and conditions set out in this Agreement.
- (D) Upon Closing and subject to the terms of this Agreement, the Issuer will hold approximately 18.00% of the total equity interest in Leju as a result of this Agreement and on the assumption that there are no changes to the issued share capital of Leju.

It is hereby agreed as follows:

1. EQUITY TRANSFER AND CONSIDERATION

- 1.1 On and subject to the terms and conditions of this Agreement, the Issuer agrees to purchase and the Sellers agrees to sell the Sale Interest, being in aggregate 24,438,564 ordinary shares and 36,687 ADSs of Leju, representing approximately 18.00% of the total equity interest in Leju as at the date of this Agreement.
- 1.2 The consideration for the sale and purchase of the Sale Interest shall be the sum of US\$93,600,000 (equivalent to approximately HK\$725,400,000, being approximately

US\$3.82 per ordinary share or ADS of Leju), which shall be settled by the Issuer allotting and issuing Consideration Shares, being 78,676,790 newly issued Shares, to the Sellers and/or the Eligible Nominee(s) (as defined below), at the issue price of HK\$9.22 (equivalent to approximately US\$1.19), on and subject to the terms and conditions set out in this Agreement.

1.3 The Consideration Shares shall, when issued, be free and clear from all Encumbrance and shall, upon Closing, be credited as fully paid and rank *pari passu* in all respects with the other Shares in issue including the rights attaching to them.

2. CONDITIONS TO CLOSING

- 2.1 Subject to the provisions under this clause 2, Closing shall be conditional on the Closing Conditions being satisfied or waived in accordance with clause 2.2 with respect to all Parties.
- 2.2 The Closing Conditions set out in <u>Schedule 2</u> paragraph 3 may be waived, in whole or in part, at any time by the Sellers, by notice in writing to the Issuer and the Closing Conditions set out in <u>Schedule 2</u> paragraph 4 may be waived, in whole or in part, at any time by the Issuer, by notice in writing to the Sellers. For the avoidance of doubt, no other Closing Conditions may be waived by any Party.
- 2.3 The date on which all the Closing Conditions set out in <u>Schedule 2</u> have been satisfied (or waived in accordance with clause 2.2) shall be the "**Unconditional Date**". If the Unconditional Date has not occurred on or prior to ninety (90) days after the date of this Agreement (the "**Longstop Date**"):
 - (a) this Agreement shall terminate with immediate effect (other than the Surviving Provisions), and (without prejudice to the rights and/or obligations of any Party in respect of any antecedent breach) the Parties shall be released and discharged from their respective obligations under this Agreement;
 - (b) except that the Parties may upon mutual written agreement, within five (5) Business Days after the Longstop Date, extend the Longstop Date to such later date as the Parties may agree (such date being not more than ninety (90) days after the Longstop Date) (such new date being the "Extended Longstop Date").
- 2.4 If the Unconditional Date has not occurred on or prior to the Extended Longstop Date (if any), this Agreement shall terminate with immediate effect (other than the Surviving Provisions and (without prejudice to the rights and/or obligations of any Party in respect of any antecedent breach) the Parties shall be released and discharged from their respective obligations under this Agreement.
- 2.5 The Issuer shall use its reasonable endeavours to ensure that the Closing Conditions set out in <u>Schedule 2</u> (excluding <u>Schedule 2</u> paragraph 4) are fulfilled or satisfied as soon as reasonably practicable after the date of this Agreement and remain so up to or as of the Closing Date (as the case may be).
- The Sellers shall use their reasonable endeavours to ensure that the Closing Conditions set out in <u>Schedule 2</u> (excluding <u>Schedule 2</u> paragraph 3) be fulfilled or satisfied as soon as reasonably practicable after the date of this Agreement and remain so up to or as of the Closing Date (as the case may be).

3. CLOSING

3.1 Subject to the satisfaction or waiver of the Closing Conditions in accordance with the provisions of this Agreement, Closing shall take place at the offices of Skadden, Arps,

Slate, Meagher & Flom, at 42nd Floor, Edinburgh Tower, The Landmark, 15 Queen's Road Central, Hong Kong, on the Closing Date at 2:00 p.m. (Hong Kong time) or by such other method (including the electronic exchange of relevant documents and items) or at such other place and time as the Parties may agree in writing.

- 3.2 At Closing, each of the Parties shall deliver or perform (or ensure that there is delivered or performed) all those documents, items and actions respectively listed in relation to that Party in <u>Schedule 3</u>.
- 3.3 All documents and items delivered at Closing pursuant to clause 3.2, shall be held by the recipient to the order of the person delivering the same until such time as Closing shall be deemed to have taken place. Simultaneously with delivery of all documents and all items required to be delivered at such Closing pursuant to clause 3.2 (or waiver of the delivery of it by the person entitled to receive the relevant document or item), the documents and items delivered in accordance with Schedule 3 shall cease to be held to the order of the person delivering them, the actions required to take place in accordance with Schedule 3 shall occur and Closing shall be deemed to have taken place.
- 3.4 Each Seller shall be entitled to nominate one or more entities to take up its Consideration Shares upon Closing by giving written notice to the Issuer no later than three (3) Business Days before the Closing Date, provided that:
 - (a) such entities are majority owned and controlled by the Seller ("Eligible Nominee(s)"); and
 - (b) such nomination shall not affect the liabilities and obligations of the Seller under this Agreement.

The written notice referred to in this clause 3.4 shall contain the full name and legal address of each relevant Eligible Nominee and the number of Consideration Shares to be taken up by it, and the full name and address of the process agent in Hong Kong of such Eligible Nominee (if applicable).

4. REPRESENTATIONS, WARRANTIES AND UNDERTAKINGS

- 4.1 Subject to and except for the information in any announcement, circular or other release or communication released or published by the Issuer, the Issuer represents and warrants to the Sellers that the Issuer's Warranties set out in Part A of Schedule 4 are true, accurate and complete as at the date of this Agreement. The Issuer's Warranties, save for those that are expressed to be given only as of a specific date, shall be deemed to be repeated by the Issuer immediately prior to Closing by reference to the facts and circumstances then existing as if references in the Issuer's Warranties to the date of this Agreement were references to the Closing Date.
- 4.2 The Sellers represent and warrant to the Issuer that the Sellers' Warranties set out in Part B of Schedule 4 are true, accurate and complete as at the date of this Agreement. The Sellers' Warranties, save for those that are expressed to be given only as of a specific date, shall be deemed to be repeated by the Sellers for themselves and on behalf of any Eligible Nominees immediately prior to Closing by reference to the facts and circumstances then existing as if references in the Sellers' Warranties to the date of this Agreement were references to the Closing Date.
- 4.3 Each Warranty given pursuant to clauses 4.1 and 4.2 is a separate and independent statement and is not limited or otherwise affected by any other Warranty or by any other provision of this Agreement.

4.4 Each Party hereby undertakes to notify the other Parties in writing promptly if it becomes aware of any circumstance arising at any time after the date of this Agreement which would cause any Warranty (if the Warranties were repeated with reference to the facts and circumstances then existing) to become untrue or inaccurate or misleading if such Warranty were repeated at any time prior to Closing.

5. EXPENSES

- 5.1 Save as otherwise expressly provided in this Agreement, each of the Issuer and the Sellers shall be liable for its own costs and expenses (including but not limited to the costs and expenses of their own legal and other professional advisers) incurred in connection with the transactions contemplated herein.
- 5.2 Without prejudice to clause 5.1, all stamp, issue, registration, documentary or other similar taxes and duties payable on or in connection with the allotment and issue of the Consideration Shares shall be paid by the Issuer.
- 5.3 The Issuer agrees to pay all fees, costs and expenses incurred in connection with the listing of the Consideration Shares on the Hong Kong Stock Exchange.
- 5.4 Without prejudice to clause 5.1, all stamp, issue, registration, documentary or other similar taxes and duties payable on or in connection with the transfer of the Sale Interest shall be paid by the Sellers.

6. INFORMATION AND REPORTING

The Sellers shall, and shall procure their Affiliates to, comply with Applicable Laws (including but not limited to the Listing Rules and the Takeovers Code) in respect of the transactions contemplated under this Agreement and supply such information in relation to itself or its Affiliates or any transferee or prospective transferee of the Consideration Shares as may be necessary for inclusion in the documents to be despatched or announcements to be issued by the Issuer in accordance with or pursuant the Listing Rules, the Takeovers Code or other Applicable Laws or as requested by any applicable regulatory authority including the Hong Kong Stock Exchange and the SFC, take responsibility for such information in relation to itself and its Affiliates and authorise the publication, despatch and/or release of such documents and announcements containing information provided by it or its Affiliates.

7. CONFIDENTIALITY

- 7.1 Each Party undertakes that it shall (and shall procure that its Affiliates shall, and where relevant, undertakes to procure that its officers, employees, agents, investment managers and professional and other advisers and those of any Affiliate (together its "Authorised Persons") shall) use its best endeavours to keep confidential at all times and not permit or cause the disclosure of any information (other than to its Authorised Persons) which it may have or acquire before or after the date of this Agreement relating to the provisions of, and negotiations leading to, this Agreement and the performance of the obligations thereunder (such information being "Confidential Information"). In performing its obligations under this clause 7.1, each Party shall apply confidentiality standards and procedures at least as stringent as those it applies generally in relation to its own confidential information.
- Each of the Parties agrees that it shall not use Confidential Information for any purpose other than in relation to the proper performance of its obligations and exercise of its rights under this Agreement and the transactions contemplated hereunder.

- 7.3 Each Party shall use its reasonable endeavours to alert the other Party as soon as is reasonably practical after it becomes aware of any request from a third party for disclosure of any Confidential Information.
- 7.4 The obligation of confidentiality under clause 7.1 does not apply to:
 - (a) information which at the date of disclosure is within the public domain (otherwise than as a result of a breach of this clause 7);
 - (b) the disclosure of information to the extent required to be disclosed by law, regulation or any court, tribunal or regulatory authority; or
 - (c) any announcement made in accordance with the terms of clause 9.

8. TERMINATION

- 8.1 If at any time prior to Closing it becomes apparent that any of the Issuer's Warranties is, or any event occurs or matter arises which renders, any of the Issuer's Warranties untrue or incorrect in any material respect as at the date of this Agreement or at Closing, then, the Sellers may, in their sole and reasonable discretion, by notice in writing to the Issuer, terminate this Agreement with immediate effect and the provisions of clause 8.4 shall apply.
- 8.2 If at any time prior to Closing it becomes apparent that any of the Sellers' Warranties is, or any event occurs or matter arises which renders, any of the Sellers' Warranties untrue or incorrect in any material respect as at the date of this Agreement or at Closing, then the Issuer may, in its sole and reasonable discretion, by notice in writing to the Sellers, terminate this Agreement with immediate effect and the provisions of clause 8.4 shall apply.
- 8.3 Without prejudice to clauses 2.3, 2.4, 8.1 or 8.2, this Agreement may be terminated at any time prior to Closing by written agreement between the
- 8.4 If there is a termination pursuant to clauses 2.3, 2.4, 8.1, 8.2 or 8.3, this Agreement shall terminate between the Parties and neither Party shall have any claim against the other in respect of any matter or thing arising out of or in connection with this Agreement except (i) in relation to obligations, agreements and liabilities arising prior to such termination, (ii) in relation to any antecedent breach of any obligation under this Agreement and (iii) the continued application of the Surviving Provisions.

9. ANNOUNCEMENTS

- 9.1 Subject to clauses 9.2 and 9.3, no public announcement of any kind in connection with this Agreement or the transactions contemplated thereunder shall be made or issued by or on behalf of any Party or any of its Affiliates without the prior written consent of the other Party.
- 9.2 If any Party is required by law or by any stock exchange or by any governmental or regulatory authority to make any announcement in connection with this Agreement or the transactions contemplated thereunder (other than the announcement referred to in clause 9.3), the relevant Party shall, to the extent legally permissible, immediately notify the other Party and shall use all reasonable endeavours to accommodate the requests of such Party with respect to the terms and provisions of such announcement.
- 9.3 The Parties acknowledge and agree that the Issuer is required to publish or despatch announcements, circulars or other communications ("Communications") in relation to the subject matter contained herein following entry into this Agreement by the Parties

in accordance with the Listing Rules. The Issuer will provide the Sellers with a draft of such Communications and permit the Sellers a reasonable period of time in all the circumstances to review and comment, and to use its reasonable efforts to accommodate all reasonable comments made by the Sellers, on the Communications prior to the publication or despatch of the Communications.

10. NOTICES

- Any notice in connection with this Agreement shall be in writing and delivered by hand, fax, e-mail, registered post or courier using an internationally recognised courier company. A notice shall be effective upon receipt and shall be deemed to have been received (i) at the time of delivery, if delivered by hand, registered post or courier, (ii) at the time of dispatch, if delivered by fax (subject to confirmation of uninterrupted transmission by the sender by a transmission report), or (iii) in the case of e-mail, at the time it leaves the email gateway of the sender, provided that, in either case, where delivery occurs outside Working Hours, notice shall be deemed to have been received at the start of Working Hours on the next following Business Day.
- 10.2 The contact details of the Parties for the purpose of clause 10.1 are:

For the Issuer:

Address: 11/F, Qiu Shi Building, No.788 Guangzhou Road, Jing'an District, Shanghai, China

2000072

E-mail Address: zhouliang@ehousechina.com

For the attention of: Fred Zhou

With a copy to: Skadden, Arps, Slate, Meagher & Flom

Address: 42F, Edinburgh Tower, The Landmark, 15 Queen's Road Central, Hong Kong

E-mail Address: christopher.betts@skadden.com

For the attention of: Mr. Christopher Betts

For the Sellers:

Address: SINA Plaza, No. 8 Courtyard 10 West Xibeiwang E. Road, Haidian District, Beijing

100193, People's Republic of China

Fax No: (8610) 82607527 E-mail Address: haiyan@staff.sina.com.cn

For the attention of: Helen Gu

All notices or formal communications under or in connection with this Agreement shall be in the English language or, if in any other language, accompanied by a translation into English. In the event of any conflict between the English text and the text in any other language, the English text shall prevail.

11. ASSIGNMENT

11.1 No Party may assign its rights under this Agreement without the prior written consent of the other Party except that the Sellers may make the nomination in accordance with clause 3.4.

12. FURTHER ASSURANCE

- 12.1 Each Party shall perform (or procure the performance of) all further acts and things and execute and deliver (or procure the execution and delivery of) such further documents, as may be required by law or as may be necessary or reasonably required by the Parties to implement and give effect to this Agreement.
- 12.2 Each Party shall procure that its Affiliates comply with all obligations under this Agreement which are expressed to apply to any such Affiliates.

13. GOVERNING LAW; JURISDICTION AND ARBITRATION

- 13.1 This Agreement is governed by and shall be construed in accordance with Hong Kong law.
- Any dispute, controversy, difference or claim arising out of, in connection with or relating to (in each such case, in any manner whatsoever) this Agreement (each a "**Dispute**"), including a Dispute as to the validity or existence of this Agreement, shall be resolved in accordance with the provisions of this clause 13.
 - (a) A Dispute shall be referred to and resolved finally by arbitration in Hong Kong administered by the Hong Kong International Arbitration Centre (the "HKIAC") in accordance with the Hong Kong International Arbitration Centre Administered Arbitration rules (the "HKIAC Rules") in force when the Notice of Arbitration is submitted in accordance with the HKIAC Rules, save as modified in clause 13.2(b) below.
 - (b) A Dispute shall be resolved by an arbitral tribunal consisting of three (3) arbitrators, one appointed by the Issuer, one appointed by the Sellers and the third appointed by the first two arbitrators. The arbitration (including arbitration proceedings and hearing) shall be conducted in English.
 - (c) Any arbitral award shall be final and binding upon the Parties thereto and shall be enforceable in accordance with its terms. Neither of the Parties shall seek to commence any judicial proceeding with a view to appealing, reviewing or setting aside any arbitral award. All such rights of appeal or judicial review of any arbitral award as would otherwise be exercisable by the Parties are hereby excluded to the fullest extent permitted.
 - (d) Any arbitral award may be enforced by filing as a judgment in any court of competent jurisdiction or by any other competent application or proceeding in any such court for the enforcement of the arbitral award, as the case may be.

14. CONTRACTS (RIGHTS OF THIRD PARTIES) ORDINANCE

A person who is not a party to this Agreement shall not have any rights under the Contracts (Rights of Third Parties) Ordinance (Chapter 623 of the Laws of Hong Kong) to enforce any terms of this Agreement and the Parties may amend, vary, waive, terminate or rescind this Agreement at any time and in any way without the consent of and without notice to any third party.

15. ENTIRE AGREEMENT

15.1 This Agreement, together with any agreements or documents referred to herein, sets out the entire agreement and understanding between the Parties with respect to its subject matter and supersedes all prior agreements, understandings, negotiations and discussions (whether oral or written) and all previous agreements in relation to the

subject matter contained herein are hereby terminated and shall have no further force or effect.

16. COUNTERPARTS

16.1 This Agreement may be executed in any number of counterparts, and by each Party on separate counterparts. Each counterpart is an original, but all counterparts shall together constitute one and the same instrument. Delivery of a counterpart of this Agreement by e-mail attachment or telecopy shall be an effective mode of delivery.

17. APPOINTMENT OF PROCESS AGENT

- 17.1 Each of the Sellers hereby irrevocably appoint Starmobile Hong Kong Limited, of Unit 1-3, 20/F, Futura Plaza, 111-113 How Ming Street, Kwun Tong, Kowloon, Hong Kong, as their agent to accept service of process in Hong Kong in any legal action or proceedings arising out of this Agreement, service upon whom shall be deemed completed whether or not forwarded to or received by it.
- 17.2 The Sellers agree to inform the Issuer in writing of any change of address of their process agent or authorised representative within twenty (20) Business Days of such change.
- 17.3 If the process agent of the Sellers ceases to be able to act as such or to have an address in Hong Kong, each of the Sellers irrevocably agrees to promptly appoint a new process agent in Hong Kong acceptable to the Issuer and to deliver to the Issuer within ten (10) Business Days a copy of a written acceptance of appointment by the process agent.
- 17.4 Nothing in this Agreement shall affect the right to serve process in any other manner permitted by law.

In Witness Whereof this Agreement has been duly executed SIGNED by	uted b)	y the authorised representatives of the Parties on the date first above written.
Zhou Xin for and on behalf of E-House (China) Enterprise Holdings Limited in the presence of: /s/ [WITNESS])) -)	/s/ Zhou Xin

SIGNED by)		
for and on behalf of Sina Corporation)	/s/ Charles Chao	
in the presence of:)		
SIGNED by)		
for and on behalf of MemeStar Limited	,))	/s/ [SIGNATORY]	
in the presence of:)		

SCHEDULE 1 SALE INTEREST AND CONSIDERATION SHARES

Seller	Ordinary shares in Leju	ADSs in Leju	Consideration Shares
Sina	24,438,564	_	_
MemeStar	_	36,687	78,676,790
	i		

SCHEDULE 2 CLOSING CONDITIONS

- 1. The approval of this Agreement and the transactions and other arrangement (as necessary) contemplated hereunder, including the acquisition the Sale Interest and allotment and issue of the Consideration Shares, by the requisite majority of shareholders or independent shareholders of the Issuer (as appropriate) at an EGM as required under Applicable Laws (including the Takeovers Code and the Listing Rules).
- 2. The granting of the approval for the listing of, and permission to deal in, the Consideration Shares by the Hong Kong Stock Exchange.
- 3. The Issuer's Warranties set out in Schedule 4 Part A continuing to be true, accurate and complete in all material respects as of the Closing Date.
- 4. The Sellers' Warranties set out in Schedule 4 Part B continuing to be true, accurate and complete in all material respects as of the Closing Date.

SCHEDULE 3 CLOSING ARRANGEMENT

Part A Issuer Obligations

At Closing or otherwise within the time period specifically stated below, the Issuer shall perform all (but not part only unless the Sellers so agree) of the following:

- 1. issue and allot the Consideration Shares to the Sellers or the relevant Eligible Nominee(s), as the case may be, credited as fully-paid, and procure the entry of the Sellers or the relevant Eligible Nominee(s), as the case may be, in the register of members of the Issuer as the owner of the relevant Consideration Shares; and
- 2. deliver or ensure that there is delivered to the Sellers:
 - (a) (i) definitive certificates in respect of the relevant Consideration Shares in favour of the Sellers and/or its Eligible Nominee(s); or (ii) deposit the certificates into the account of the relevant CCASS Participant with whom the Seller has accounts in accordance with the Seller's instructions;
 - (b) a certified copy or extract of the resolutions of the Board with respect to approval of (i) the execution of this Agreement and the performance of the Issuer's obligations hereunder; and (ii) the issue of Consideration Shares to the Sellers or the relevant Eligible Nominee(s), as the case may be, in accordance with this Agreement; and
 - (c) a certificate signed by a duly authorised officer of the Issuer certifying that the matters in Schedule 2 paragraph 3 are true and correct to the Sellers.

Part B Seller's Obligations

At Closing, the Sellers shall:

- 1. deliver or ensure that there is delivered to the Issuer:
 - (a) a duly executed share transfer form in respect of the relevant Sale Interest in favour of the Issuer together with the share certificates relating to such Sale Interest (or an indemnity in respect of any lost unavailable share certificates in the agreed terms);
 - (b) a certified copy or extract of the resolutions of the board of MemeStar with respect to approval of the execution of this Agreement and the performance of its respective obligations hereunder; and
 - (c) a certificate signed by a duly authorised officer of the Sellers certifying that the matters in Schedule 2 paragraph 4 are true and correct for itself and behalf of any Eligible Nominees to the Issuer.

SCHEDULE 4 REPRESENTATIONS AND WARRANTIES

Part A Issuer's Warranties

The Issuer represents and warrants to the Sellers that:

- 1. **Incorporation and Subsidiaries:** each member of the Group is a company duly incorporated and validly existing and having its capital fully paid up under the laws of its jurisdiction of incorporation, is in compliance with its constitutional documents, and all laws, regulations and corporate governance requirements to which it is subject, is not in liquidation or receivership, has full power and authority to own its properties and to conduct its business and is lawfully qualified and licensed to do business in those jurisdictions in which business is conducted by it, and each of the foregoing statements in this paragraph is true with respect to each of the branches established by a member of the Group;
- 2. **Validity of Agreement:** this Agreement has, subject to the satisfaction of the Closing Conditions, been duly authorised, executed and delivered by the Issuer and constitutes valid and legally binding obligations of the Issuer, enforceable against the Issuer in accordance with its terms;
- 3. **Consideration Shares:** the Consideration Shares when issued and delivered:
 - (a) will be duly and validly issued and fully-paid and non-assessable;
 - (b) will rank *pari passu* with, and carry the same rights in all respects as the other Shares in issue and shall be entitled to all dividends and other distributions declared, paid or made thereon the record date of which falls after the Closing Date;
 - (c) will not be subject to any kind of pre-emptive rights;
 - (d) will be freely transferable, free and clear of all liens, charges, encumbrances, security interests or claims of third parties and will not be subject to calls for further funds; and
 - (e) will be duly listed, and admitted to trading, on the Main Board of the Hong Kong Stock Exchange;
- 4. **Restrictions:** save as otherwise provided for under this Agreement and in the Articles:
 - (a) there are no restrictions on transfers of the Consideration Shares; and
 - (b) there are no restrictions on the voting or transfer of any of the Consideration Shares or payments of dividends with respect to the Consideration Shares pursuant to the Issuer's constitutional documents, or pursuant to any agreement or other instrument to which the Issuer is a party or by which it is bound;
- 5. **Listing**: all of the issued Shares have been duly listed on the Hong Kong Stock Exchange;
- 6. **Consents:** save for the matters set out in the Closing Conditions, no consent, clearance, approval, authorisation, order, registration or qualification of or with any court, governmental agency or regulatory or self-regulatory body having jurisdiction over the Issuer is required and no other action or thing is required to be taken, fulfilled or done (including without limitation the obtaining of any consent or licence or the making of any filing or registration) for the execution and delivery by the Issuer of the Agreement, the issue of the Consideration Shares to the Sellers, the carrying out of the other transactions contemplated by this Agreement except for all of those which have been,

or will prior to the Closing Date be, obtained and are, or will on the Closing Date be, in full force and effect. The Issuer has no knowledge of any facts that would reasonably lead to delisting or suspension of the Shares in the foreseeable future and the issuance by the Issuer of the Consideration Shares shall not have the effect of delisting or suspending the Shares from the Hong Kong Stock Exchange;

- 7. **Compliance:** the execution, delivery and performance of the Agreement, the issue of the Consideration Shares, the carrying out of the other transactions contemplated by the Agreement do not and will not:
 - (a) conflict with or result in a breach of any of the terms or provisions of, or constitute a default (nor has any event occurred which, with the giving of notice and/or the passage of time and/or the fulfilment of any other requirement would result in a default) by the Issuer or any other Group Company under, (a) the documents constituting the Issuer, or (b) any indenture, contract, lease, mortgage, deed of trust, note agreement, loan agreement or other agreement, obligation, condition, covenant or instrument to which the Issuer or any other Group Company is a party or by which any of their respective assets are bound; or
 - (b) infringe any existing applicable law, rule, regulation, judgment, order, authorisation or decree of any government, governmental or regulatory body or court, domestic or foreign, having jurisdiction over the Issuer or any other Group Company or any of their respective assets; or
 - (c) infringe the Listing Rules;
- 8. **No Suspension:** since the listing of the Shares on the Main Board of the Hong Kong Stock Exchange on 20 July 2018, and save for any trading halt required under the Listing Rules or the Takeovers Code in connection with the transactions contemplated herein, (i) the Shares have been listed on the Hong Kong Stock Exchange, (ii) trading in the Shares has not been suspended by the Hong Kong Stock Exchange since the listing on the Hong Kong Stock Exchange, and (iii) the Issuer has received no communication, written or oral, from the Hong Kong Stock Exchange regarding the suspension or delisting of the Shares from the Hong Kong Stock Exchange;
- 9. **Encumbrances:** no encumbrancer has taken possession of, and no receiver has been appointed over, the whole or any substantial part of the assets or undertaking of the Issuer or any other Group Company;
- Bankruptcy: neither the Issuer nor any of its Subsidiaries is insolvent or bankrupt or unable to pay its debts, has not stopped, suspended or threatened (through actions of its board of directors or otherwise) to stop or suspend payment of all or any part of (or of a particular type of) its debts, has not proposed or made any agreement for the deferral, rescheduling or other readjustment of all of (or of a particular type of) its debts (or of any part which it will or might otherwise be unable to pay when due), has not proposed or made a general assignment or an arrangement or composition with or for the benefit of the relevant creditors in respect of any such debts and a moratorium has not been agreed or declared in respect of or affecting all or any part of (or of a particular type of) its debts; and
- 11. **Bankruptcy Proceedings:** no order has been made and no effective resolution has been passed for the winding-up or dissolution or administration or receivership of the Issuer or any of the Group Companies, and neither the Issuer nor any of the Group Companies has threatened (through actions of its board of directors or otherwise) to cease to carry on all or a material part of its business or operations except to the extent that it has made disposals of assets or shares in the ordinary course of its business for fair value.

12. **Disclosure**: the Issuer has fully disclosed all material information in accordance with the Listing Rules and such disclosures are true, correct and complete in all material respects. No such disclosures contain any untrue statement of a material fact or omits to state a material fact necessary in order to make the statements contained therein not misleading in light of the circumstances under which they were made.

Part B Seller's Warranties

The Sellers represents and warrants to the Issuer that:

- 1. with respect to itself and any Eligible Nominees:
 - (a) it is a company duly incorporated and organised and validly existing under the laws of its jurisdiction of incorporation or organisation, is not in liquidation or receivership and has full power and authority to own its properties and to conduct its business;
 - (b) it has power under its constitutional documents to subscribe for the Consideration Shares and hold the Consideration Shares upon the terms set out herein;
 - (c) it has full power and authority to enter into this Agreement and to perform its obligations hereunder;
 - (d) it is an independent third party and to the best knowledge of the Sellers after making reasonable enquiries it is not a connected person or an associate (each as defined in the Listing Rules) of the Issuer or any of its Subsidiaries or their respective directors, chief executives or substantial shareholders (as defined in the Listing Rules). For the avoidance of doubt it is noted upon Closing (as contemplated under this Agreement), both Sina and the Issuer will become shareholders of Leju;
 - (e) this Agreement has been duly authorised, executed and delivered by the Sellers and constitutes valid and legally binding obligations of the Sellers, enforceable in accordance with their respective terms; and
 - (f) the particulars and information relating to it as set out in this Agreement are true, correct and accurate and any information provided to the Issuer for the purposes of any Communications was, or will be, true, correct and accurate.

2. **Sale Interest:** the Sale Interest:

- (a) are beneficially owned by the respective Sellers;
- (b) when delivered:
 - (i) will be duly and validly issued and fully-paid and non-assessable;
 - (ii) will rank *pari passu* with, and carry the same rights in all respects as the other Leju ordinary shares and/or Leju ADSs (as the case may be) in issue and shall be entitled to all dividends and other distributions declared, paid or made thereon the record date of which falls after the Closing Date;
 - (iii) will not be subject to any kind of pre-emptive rights;

- (iv) will be freely transferable, free and clear of all liens, charges, encumbrances, security interests or claims of third parties and will not be subject to calls for further funds; and
- (c) are duly listed, and admitted to trading, on the Nasdaq Global Select Market;
- 3. **Restrictions:** save as otherwise provided for under this Agreement and in the articles of the Sellers:
 - (a) there are no restrictions on transfers of the Sale Interest; and
 - (b) there are no restrictions on the voting or transfer of any of the Sale Interest or payments of dividends with respect to the Sale Interest pursuant to Leju's constitutional documents, or pursuant to any agreement or other instrument to which any Seller is a party or by which it is bound;
- 4. **Listing**: all of the Sale Interest have been duly listed on the Nasdaq Global Select Market.

SCHEDULE 5 DEFINITIONS

- The following terms and expressions used in this Agreement, unless the context otherwise requires, shall have the following meanings:
 - "Affiliate" of any specified person means any other person directly or indirectly Controlling or Controlled by or under direct or indirect common Control with such specified person;
 - "Applicable Laws" means, with respect to any person, any laws, rules, regulations, directives, treaties, decrees or orders of any governmental or regulatory authority that are applicable to and binding on such person and without limitation to the foregoing, shall, in respect of the Issuer, include the Listing Rules and the Takeovers Code;
 - "Articles" means the Issuer's articles of association from time to time;
 - "**Authorised Persons**" has the meaning given to it in clause 7.1;
 - "Board" means the board of directors of the Issuer;
 - "Business Day" means a day (other than a Saturday or Sunday or public holiday and any other day on which a tropical cyclone warning no. 8 or above or a "black" rain warning signal is hoisted in Hong Kong) on which commercial banks are open for business in the city in which the specified office of the registrar is located and in Hong Kong, and "Business Days" shall be construed accordingly;
 - "CCASS" means the Central Clearing and Settlement System established and operated by Hong Kong Securities Clearing Company Limited;
 - "CCASS Participant" means, as regards any Shares deposited with CCASS, the person admitted by HKSCC as a Participant (as defined in the General Rules of CCASS, as amended, issued by HKSCC);
 - "Closing" means the completion of the acquisition of the Sale Interest, as the case may be, in accordance with clause 3;
 - "Closing Conditions" means the conditions to Closing as set out in Schedule 2;
 - "Closing Date" means the date of Closing, which shall be a date not later than the fifth (5th) Business Day after the Unconditional Date, or such other date as the applicable Parties may agree in writing;
 - "Communications" has the meaning given to it in clause 9.3;
 - "Confidential Information" has the meaning given to it in clause 7.1;
 - "Consideration Shares" has the meaning given to it in Recital (B);
 - "Control" means in relation to an undertaking:
 - (a) the power to direct the exercise of a majority of the voting rights capable of being exercised at a general meeting of that undertaking;
 - (b) the right to appoint or remove a majority of the board of directors (or corresponding officers) of that undertaking; or
 - (c) the right to exercise a dominant influence over that undertaking by virtue of provisions contained in its constitutional documents or under a control contract or otherwise.

In each case either directly or indirectly and "Controlled" and "Controlling" shall be construed accordingly;

- "Director" means a director of the Issuer;
- "Dispute" has the meaning given to it in clause 13.2;
- "EGM" means the special general meeting of the Issuer to be held to approve, among other things, the acquisition of the Sale Interest and the issue of the Consideration Shares (as required under the Listing Rules and the Takeovers Code);
- "Eligible Nominee(s)" has the meaning given to it in clause 3.4(a);
- "Encumbrance" means a mortgage, charge (fixed or floating), pledge, lien, option, restriction, right of first refusal, right of pre-emption, debenture, assignment by way of security, trust arrangement for the purpose of providing security, third-party right or interest, other encumbrance or security interest of any kind, or another type of preferential arrangement (including, without limitation, a title transfer or retention arrangement) having similar effect, and any agreement to create any of the foregoing;
- "Extended Longstop Date" has the meaning given to it in clause 2.3(b);
- "Fairly Disclosed" means fairly and adequately disclosed in a manner and with sufficient details to enable the Sellers to reasonably identify and understand the nature and scope of the matter being disclosed to qualify the Issuer's Warranties;
- "Group" means the Issuer and together with its Subsidiaries from time to time;
- "Group Companies" means members of the Group, and "Group Company" means any one of them;
- "HK\$" means the legal currency of Hong Kong;
- "**HKIAC**" has the meaning given to it in clause 13.2(a);
- "HKIAC Rules" has the meaning given to it in clause 13.2(a);
- "HKSCC" means Hong Kong Securities Clearing Company Limited, a wholly-owned subsidiary of Hong Kong Exchanges and Clearing Limited;
- "Hong Kong" means the Hong Kong Special Administrative Region of the People's Republic of China;
- "Hong Kong Stock Exchange" means The Stock Exchange of Hong Kong Limited;
- "Issuer's Warranties" means the warranties set out in Schedule 4 Part A;
- "Listing Rules" means the Rules Governing the Listing of Securities on The Stock Exchange of Hong Kong Limited;
- "Longstop Date" has the meaning given to it in clause 2.3;
- "SFC" means the Securities and Futures Commission of Hong Kong;
- "Shares" means the ordinary shares with a par value of USD0.00001 each in the share capital of the Issuer;
- "Seller's Warranties" means the warranties set out in Schedule 4 Part B;
- "Subsidiary" includes, in relation to any person: (i) any company or business entity of which that person owns or controls (either directly or through one or more other subsidiaries) more than 50 per cent. of the issued share capital or other ownership interest having ordinary voting power to elect the directors, managers or trustees of such company or business entity; (ii) any company or business entity of which that person owns or controls (either directly or through one or more other subsidiaries) not more than 50 per cent. of the issued share capital or other ownership interest having ordinary voting power to elect directors, managers or trustees of such company or business entity but effectively controls (either directly or through one or more other subsidiaries) the management of the direction of business operations of such company or business entity; and (iii) any company or business entity which at any time has its accounts consolidated with those of that person or which, under Hong Kong law or any other applicable law regulations or the or such other applicably generally accepted accounting principles from time to time, should have its accounts consolidated with those of that person;

- "Surviving Provisions" means clauses 1.3, 7, 10 and 13;
- "Takeovers Code" means the Codes on Takeovers and Mergers and Share Buy-backs;
- "Unconditional Date" has the meaning given to it in clause 2.3;
- "USD" means United States dollars, the lawful currency of the United States;
- "Warranties" mean the Issuer's Warranties and Seller's Warranties as set out in Schedule 4;
- "Working Hour" means 9.30 a.m. to 5.30 p.m. on a Business Day in the place of receipt of a notice.

2. In this Agreement:

- (a) words denoting the singular shall include the plural and vice versa;
- (b) words denoting one gender shall include each gender and all genders;
- (c) references to persons shall be deemed to include references to natural persons, to firms, to partnerships, to bodies corporate, to undertakings, to associations, to organisations, to trusts, to trustees, to legal representatives, to governments (or any department or agency thereof) or to any other entity howsoever designated or constituted (in each case, whether or not having separate legal personality), but references to individuals shall be deemed to be references to natural persons only;
- (d) a reference to this Agreement includes the Schedules to it, each of which forms part of this Agreement for all purposes;
- (e) references to clauses and Schedules are, unless stated otherwise, references to clauses and schedules of this Agreement;
- (f) references to paragraphs are, unless expressly provided otherwise, references to paragraphs of the Schedule in which the references appear;
- (g) the headings are inserted for convenience only and will not affect the construction of this Agreement;
- (h) any reference to an enactment or a statutory provision is a reference to it as it may have been or may from time to time be, amended, modified, consolidated or re-enacted;
- (i) the terms "hereof" and "hereunder" (and any other similar expressions) refer to this Agreement and not to any particular clause or other portion hereof and include any agreement supplemental hereto; and

(j) any statement in this Agreement qualified by the expression "to the best knowledge of the Issuer or any member of the Group" or "so far as the Issuer is aware" or any similar expression shall be a reference to the actual knowledge of the Directors and the members of the senior management team of the Issuer and shall be deemed to include an additional statement that it has been made after due and careful enquiry by such persons.